Author's response to reviews

Title: Naturopaths practice behaviour: provision and access to information on complementary and alternative medicines

Authors:

Caroline A Smith (caroline.smith@unisa.edu.au)
Karen L Martin (karen.martin@unisa.edu.au)
Elizabth Hotham (Libby.Hotham@unisa.edu.au)
Susan Semple (Susan.Semple@unisa.edu.au)
Geraldine Bloustien (Gerry.Bloustien@unisa.edu.au)
Deepa Rao (Deepa.Rao@unisa.edu.au)

Version: 3 Date: 8 July 2005

Author's response to reviews: see over
Friday, 8 July 2005.

Dear Dr Dennison,

**Title:** Naturopaths practice behaviour: provision and access to information on complementary and alternative medicine. MS 1342622647648957.

Thank you for your email with news of the anticipated acceptance of this paper.

The page breaks have been removed and the subscript removed from the references. A final proof read has been made by members of the team. We look forward to hearing from you in due course.

Kind regards

Caroline Smith